Author’s response to reviews

Title: Identification of ambiguities in the 1994 chronic fatigue syndrome research case definition and recommendations for resolution

Authors:

Dr William C Reeves (wcr1@cdc.gov)
Andrew Lloyd (A.Lloyd@unsw.edu.au)
Dr Suzanne D Vernon (svernon@cdc.gov)
Dr Nancy Klimas (Nancy.Klimas@med.va.gov)
Leonard Jason (ljason@wppost.depaul.edu)
Gijs Bleijenberg (G.Bleijenberg@cksmps.azn.nl)
Birgitta Evengard (birgitta.evengard@infect.hs.sll.se)
Peter D White (p.d.white@mds.qmw.ac.uk)
Dr Rosane Nisenbaum (ran7@cdc.gov)
Elizabeth R Unger (eru0@cdc.gov)

Version: 5 Date: 10 Dec 2003

We have added text to the Background to satisfy the reviewers request. We did not add any additional references.